Author's response to reviews

Title: Approaches to interim analysis of cancer randomised clinical trials: a survey from the Italian National Monitoring Centre for Clinical Trials.

Authors:

Irene Floriani (floriani@marionegri.it)
Nicole Rotmensz (nicole.rotmensz@ieo.it)
Elena Albertazzi (elena.albertazzi@ieo.it)
Valter Torri (torri@marionegri.it)
Marisa De Rosa (m.derosa@cineca.it)
Filippo de Braud (filippo.debraud@ieo.it)
Carlo Tomino (c.tomino@aifa.gov.it)

Version: 2 Date: 18 December 2007

Author's response to reviews:

we have added the name of one authors carlo tomino, md, which was omitted by mistake in the original version